Author’s response to reviews

Title: Effect of oral functional training on immunological abilities of older people: a case control study

Authors:

Mitsue Sato (mitsue.sato@tohto.ac.jp; mshrsgmt@gmail.com)
Masahiro Sugimoto (msugi@sfc.keio.ac.jp)
Yuko Yamamoto (yamamoto.yuko@kdu.ac.jp)
Juri Saruta (saruta@kdu.ac.jp)
Keiichi Tsukinoki (tsukinoki@kdu.ac.jp)

Version: 4 Date: 07 Dec 2017

Author’s response to reviews:

1. In your “ethical approval and consent to participate” section please confirm whether written consent was obtained from participants. If written consent was not obtained, please explain why not.

According to the reviewer’s comments, we added “Written informed consent was obtained from each subject before participating in the study.” at lines 135-136.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

According to the reviewer’s comments, we uploaded clean versions with separate figure files.